Author’s response to reviews

Title: Wound healing activities of different extracts of Centella asiatica in incision and burn wound models: an experimental animal study

Authors:

Juraiporn Somboonwong (juraiporn.s@md.chula.ac.th)
Mattana Kankaisre (simattana@mahidol.ac.th)
Boonyong Tantisira (t.boonyong@yahoo.com)
Mayuree H Tantisira (Mayuree.t@chula.ac.th)

Version: 4 Date: 10 June 2012

Author’s response to reviews:

Dear Editor,

MS: 1308506478656510
Wound healing activities of different extracts of Centella asiatica in incision and burn wound models: an experimental animal study
Juraiporn Somboonwong, Mattana Kankaisre, Boonyong Tantisira and Mayuree H Tantisira
I would like to give a new point-by-point response to your concerns. These are detailed below:
- “While we appreciate that you have made every effort to copyedit your manuscript, there are still a number of grammatical errors that preclude its publication. We would therefore recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. “
I have Dr. Ian S. Haworth (University of South California) copyedit the manuscript. We have already given special thanks to him in acknowledgment section.
- “Thank you for confirming to us that animal experiments in your study were carried out in accordance with ARRIVE guidelines, and that all care was taken to minimise the suffering of the animals. However, we would ask that you include this information in the methods section of your manuscript. “
I have already done as requested.
Any further changes that I make to my manuscript are made in blue color. Thank you very much in advance for processing my manuscript further.
Regards,
Dr.Juraiporn Somboonwong